Author’s response to reviews

Title: A semi-parametric mixed models for longitudinally measured fasting blood sugar level of adult diabetic patients

Authors:
Tafere Tilahun Aniley (tafere.tilahun@ju.edu.et;tafere27@yahoo.com)
Legesse kassa Debusho (debuslk@unisa.ac.za)
Zelalem Mehari Nigusie (zelalem2011@gmail.com)
Wondwossen Kassahun Yimer (wyimer@umc.edu)
Belay Birliie Yimer (belaya.birliie@gmail.com)

Version: 2 Date: 03 Dec 2018

Author’s response to reviews:

Dear Editor,

Again we would like to thank for your genuine comments on this final version of our manuscript. We have incorporated all the comments given and this is the final version of our manuscript with the following points:

- We have changed the title for the first section from Introduction to Background.

- We have reduced the textual overlap in the background section as commented.

- We have amended the headings in the main text to: Background, Methods, Results, Discussion, and Conclusions, so that Result and Discussion sections are separated.

- The figure legends are moved to the backmatter part of the paper next to the Reference section.

- The availability of the data sets applied to this study will be provided on the reasonable request from the author.

Finally, The authors are very greatfull for giving us valuable and constructive comments, so that we have edited and corrected many things of our manuscript. Thank you indeed.